Author's response to reviews

Title: The cytotoxicity of QMixTM and sodium hypochlorite endodontic irrigating solution on human bone marrow mesenchymal stem cells

Authors:

Ahmad AlKahtani (ksucod@gmail.com)
Sarah M Alkahtany (sweetykb@gmail.com)
Amer Mahmood (drsanil@outlook.com)
Mona Ahmed Elsafadi, (vpriya.anil@gmail.com)
Abdullah Mohammed Aldahmash (asukumaran@ksu.edu.sa)
Sukumaran Anil (drsanil@gmail.com)

Version: 2 Date: 19 February 2014

Author's response to reviews:

All suggested modifications are included in the manuscript.